209

Open Access Full Text Article

# Patient Characteristics, Management, and Outcomes of Adult Asthma in a Singapore Population: Data from the SDG-CARE Asthma Registry

Mariko Siyue Koh<sup>1,2</sup>, Sean Shao Wei Lam<sup>3-6</sup>, Xiaomeng Xu<sup>57</sup>, Jun Tian Wu<sup>3,4,8</sup>, Priyan Ratnasingham<sup>7</sup>, Ricco Marsel<sup>7</sup>, Marcus Eng Hock Ong<sup>3,9</sup>, David Bruce Matchar<sup>2,10,11</sup>, Ngiap Chuan Tan<sup>2,12</sup>, Chian Min Loo<sup>1,2</sup>

<sup>1</sup>Department of Respiratory and Critical Care Medicine, Singapore General Hospital, Singapore, Singapore, <sup>2</sup>Duke-NUS Medical School, National University of Singapore, Singapore, Singapore; <sup>3</sup>Health Services and Systems Research, Duke-NUS Medical School, Singapore, Singapore; <sup>4</sup>Health Services Research Centre, SingHealth, Singapore, Singapore; <sup>5</sup>Health Services Research Institute, SingHealth Duke-NUS Academic Medical Centre, Singapore; <sup>6</sup>Lee Kong Chian School of Business, Singapore Management University, Singapore, Singapore; <sup>7</sup>Medical, GSK, Singapore, Singapore; <sup>8</sup>NUS Yong Loo Lin School of Medicine, Singapore, Singapore; <sup>9</sup>Department of Emergency Medicine, Singapore General Hospital, Singapore, Singapore; <sup>10</sup>Department of Internal Medicine (General Internal Medicine), Duke University Medical School, Durham, NC, USA; <sup>11</sup>Department of Internal Medicine, Singapore, Sing

Correspondence: Mariko Siyue Koh, Department of Respiratory and Critical Care Medicine, Singapore General Hospital, 20 College Road, Academia Level 3, Singapore, 169856, Singapore, Tel +65 65767862, Fax +65 62275247, Email mariko.koh.s.y@singhealth.com.sg

**Purpose:** Patients with asthma in Singapore often have complex patient journeys, with diagnosis and management across various primary and speciality care settings. Real-world population health data is needed to identify care gaps and inform policies.

**Patients and Methods:** This retrospective, longitudinal cohort study assessed real-world data from adults (aged  $\geq 18$  years) with asthma in the SingHealth Chronic Obstructive Pulmonary Disease and Asthma Data Mart, an integrated database of electronic medical records of patients who attended primary and/or speciality care clinics in the SingHealth Regional Health System 01/01/2015–12/31/2020. Patients were indexed by first asthma diagnosis and categorized into cohorts of index year. Patient characteristics, asthma management and outcomes were described during baseline (1-year pre-index) and follow-up periods (1-year post-index).

**Results:** Overall, 21,215 patients were included across 4 cohorts: 2016, N=12,947; 2017, N=3419; 2018, N=2816; 2019, N=2033. Most common baseline asthma medication changed from inhaled corticosteroids (ICS) alone in the 2016 cohort (32.8% [n=4252]) to ICS/long-acting  $\beta_2$ -agonist in the 2019 cohort (33.3% [n=677]). Asthma symptom control (mean [SD] Asthma Control Test scores) improved from 2016 to 2019 during baseline (18.38 [4.93] vs 19.87 [4.56]; *p*<0.001) and follow-up (18.34 [4.23] vs 21.07 [3.51]; *p*<0.001). Mean (standard deviation [SD]) number of exacerbations per patient during follow-up decreased from 2016 to 2019 (1.91 [3.11] vs 0.89 [2.07]; *p*<0.001). Mean (SD) number of emergency department visits per patient during follow-up decreased from 0.21 (0.75) in 2016 to 0.17–0.18 (0.60–0.65; *p*<0.001) between 2017 and 2019.

**Conclusion:** Health status at first asthma diagnosis improved for each succeeding cohort from 2016 to 2019, along with improvements in patient management and outcomes. This reflects greater awareness of the condition and improved use of medication and referrals in recent years, suggesting policy changes and their implementation, including promotion of disease awareness and adoption of guideline recommendations, may improve asthma outcomes in Singapore.

**Plain Language Summary:** In Singapore, up to 4 in 100 people are living with asthma. People with asthma can experience frequent symptoms and attacks that affect day-to-day activities, create emotional stress and reduce quality of life. Asthma has a financial impact on patients and the economy, as it can lead to missed work and reduced productivity. In Singapore, managing asthma can be complex, with patients often seeing both primary care physicians and specialists from diagnosis to treatment. This study assessed trends in the care that patients with asthma in Singapore received and their outcomes over a 4-year period. Patients' medical records from primary

© 0.24 Koh et al. This work is published and licensed by Dove Medical Press Limited. The full terms of this license are available at https://www.dovepress.com/terms.ph you hereby accept the firms. Non-commercial uses of the work are permitted without any further permission from Dove Medical Press Limited, provided the work is properly attributed. For permission for commercial use of this work, please see paragraphs 4.2 and 5 of our Terms (http://www.dovepress.com/terms.ph). and specialist care were collected for 1 year before and after their asthma diagnosis to determine whether treatment changed over time, whether symptoms improved and whether use of healthcare services changed.

Between 2016 and 2019, patients with asthma had fewer, less severe asthma attacks, and fewer emergency department visits over time. The treatment types also changed; in 2019 versus 2016, more patients received combination inhalers of corticosteroids with a long-acting beta 2 ( $\beta_2$ ) agonist, or other add-on treatments, which improve patients' asthma, including symptoms. This indicates that there was an increasing trend for physicians following new guidelines for recommended asthma care and these changes in practice ensured combination inhalers were made available to patients by the government, leading to improvement in asthma outcomes.

This study shows that asthma management in Singapore improved over the 4-year period, potentially through healthcare professionals taking on board improved treatment guideline recommendations and policy changes.

Keywords: asthma, asthma control, exacerbations, real world, Singapore, health outcomes

#### Introduction

Asthma affects approximately 262 million people worldwide, corresponding to a global prevalence of 3.4%.<sup>1</sup> In Singapore, the prevalence of asthma is estimated to be 2.6–3.9% in the overall population,<sup>1–3</sup> with higher prevalence (6.3%) reported among those with Chinese ethnicity.<sup>4</sup> Patients with asthma in Singapore experience substantial disease burden, including frequent symptoms, high levels of rescue medication use, absenteeism from work/school, and exacerbations.<sup>5</sup> In particular, patients with severe asthma report high healthcare resource utilization (HCRU) and oral corticosteroid (OCS) use, and poor health-related quality of life.<sup>2,6</sup> The economic burden of asthma in Singapore is considerable, with total annual costs for adults estimated at 1.25 billion United States dollars (US\$) in 2020, with 79% of costs due to lost productivity.<sup>7</sup> Asthma control status is also associated with individual economic burden; among employed patients, annual direct and indirect costs range from \$5,230 for patients with well-controlled asthma to \$25,630 for uncontrolled asthma.<sup>7</sup>

The patient journey for individuals with asthma in Singapore is complex; patients can choose their healthcare provider, with options including publicly-funded primary care polyclinics, private primary care clinics, and publicly-and/or privately-funded hospitals and speciality care clinics.<sup>8</sup> As such, diagnosis and disease management often occur across different care settings, resulting in limited real-world information regarding patients' healthcare data. Consequently, the SingHealth-Duke-GSK Chronic Obstructive Pulmonary Disease (COPD) and Asthma Real-World Evidence (SDG-CARE) collaboration was formed in 2017, with the aim of improving use and accessibility of real-world healthcare data in Singapore.<sup>9</sup> The collaboration led to development of the SingHealth COPD and Asthma Data Mart (SCDM), an integrated database comprising real-world data for patients with asthma and/or COPD attending primary and/or speciality care clinics in the SingHealth Regional Health System, the largest public healthcare system in Singapore.<sup>9</sup> As of 2019, the SCDM included 36,047 unique patients with asthma and COPD across SingHealth's public primary and specialist care clinics.<sup>9</sup> Given asthma prevalence of 2.6–3.9%<sup>1–3</sup> in a population of ~5.7 million in 2019,<sup>10</sup> the SCDM captures approximately 20% of people living with asthma in Singapore. The SCDM records patients' demographic, clinical, laboratory and radiology data, which are updated every 24 hours, ensuring near real-time, clinically relevant data.<sup>9</sup>

Given the burden of asthma in Singapore, there is a need to identify and address gaps in asthma management. Analyzing longitudinal trends in patient characteristics and outcomes from clinical visits may help inform and improve the delivery of asthma care. As such, the objective of this study was to describe the patient characteristics of an adult asthma population, along with places of care and healthcare outcomes over a 4-year period (2016–2019), using real-world primary and speciality care data collected in the SCDM.

### Methods

#### Study Design and Data Source

This retrospective cohort study included real-world data from adults with asthma in the SCDM from January 01, 2015 to December 31, 2020 (Supplementary Figure 1). The SCDM contained electronic medical records (EMRs) of patients who

attended a SingHealth Polyclinic (SHP; primary care) and/or Singapore General Hospital respiratory medicine specialist outpatient clinic (SGH-SOC; specialty care). Patients who attended an SHP could be referred to SGH-SOCs or other speciality care clinics, while patients who attended SGH-SOCs could be referred from primary care clinics other than SHPs. Some patients included in the SCDM attended both SHPs and SGH-SOCs. Further information about the SCDM has been reported elsewhere.<sup>9</sup>

The first recorded diagnosis of asthma during the identification period (January 01, 2016 to December 31, 2019) was defined as the index event; incident patients were confirmed by the absence of an asthma diagnosis or prescription for asthma in the year prior to the date of the index event (baseline period). Asthma management was assessed at baseline (ie the index date or closest to index date) and in the 1-year follow-up period after index.

### Study Population

Eligible patients were aged  $\geq 18$  years at index with a diagnosis of asthma (301485011), asthma-COPD overlap (ACO; 3046475015) or severe asthma (1208972017) by Systematized Nomenclature of Medicine codes<sup>11</sup> recorded in the SCDM during the study period and had  $\geq 1$  year of follow-up data. Patients were excluded if they had  $\leq 1$  asthma-related visit, to ensure that patients included were routinely monitored in primary or speciality care settings.

### **Study Outcomes**

Patients were categorized into 4 cohorts based on index year, from 2016 to 2019. Full details of study variables are provided in <u>Supplementary Table 1</u>. Briefly, patients' demographic and clinical characteristics (including age, sex, race, Global Initiative for Asthma [GINA] Treatment Step, smoking status, and body mass index), and asthma medication use were assessed at the index date or the closest date after index with available data. Definitions for GINA Steps were derived from the 2015 GINA report,<sup>12</sup> and are summarized in <u>Supplementary Table 2</u>. Comorbidity data were collected during the study period and identified by the presence of International Statistical Classification of Diseases and Related Health Problems –  $10^{\text{th}}$  revision codes.<sup>13</sup>

Blood eosinophil (EOS) count and asthma symptom control status (Asthma Control Test [ACT]) data were collected during the baseline and follow-up periods. In addition, data for asthma-related HCRU (outpatient and emergency department [ED] visits), documentation of vaccinations (influenza, pneumococcal PCV13, pneumococcal PPSV23), asthma counselling and written asthma action plans (WAAPs), and asthma exacerbations were assessed during the follow-up period. Definitions for asthma exacerbations are presented in <u>Supplementary Table 3</u>. To mitigate any bias in quantifying the number of exacerbations partly based on OCS burst prescriptions, we discounted one OCS prescription per patient to account for prescriptions for standby use, and only OCS prescriptions lasting 5–7 days were considered as burst treatment for an asthma exacerbation.

# Statistical Analyses

Statistical analyses were descriptive, and patients were included in cohorts by index year. Continuous variables were reported using mean and standard deviation (SD); statistical significance was assessed using one-way analysis of variance (ANOVA). Categorical variables were reported using relative frequencies and percentages; statistical significance was assessed using Chi-square tests. Statistical significance was set at p<0.05 for all analyses. All statistical analyses were performed with STATA statistical software (StataCorp. 2017. Stata Statistical Software: Release 17.0. College Station, TX: StataCorp LLC).

# Results

### Study Population

Overall, 36,405 patients were identified in the SCDM with a total of 192,832 unique asthma-related visits (Supplementary Figure 2). After applying exclusion criteria, the final study population included 21,215 adults with asthma. Baseline demographic and clinical characteristics are presented in Table 1. The proportion of patients aged  $\geq$ 54 years was highest in the 2016 cohort (57.7% [n=7468]); this decreased to 37.6% (n=1286) in the 2017 cohort, followed

	2016 (N=12,947)	2017 (N=3419)	2018 (N=2816)	2019 (N=2033)	p value
Age (year), n (%)					
≥ 8–<34	2381 (18.4)	1119 (32.7)	836 (29.7)	559 (27.5)	<0.001
≥34–<54	3098 (23.9)	1014 (29.7)	847 (30.1)	606 (29.8)	
≥54–<80	6548 (50.6)	1175 (34.4)	1012 (35.9)	781 (38.4)	
≥80	920 (7.1)	(3.3)	121 (4.3)	87 (4.3)	
<b>Male</b> , n (%)	5531 (42.7)	1626 (47.6)	1326 (47.1)	947 (46.6)	<0.001
<b>Race</b> , n (%)					
Chinese	7039 (54.4)	1645 (48.1)	1389 (49.3)	1095 (53.9)	<0.001
Malay	3042 (23.5)	947 (27.7)	760 (27.0)	525 (25.8)	
Indian	1865 (14.4)	507 (14.8)	401 (14.2)	232 (11.4)	
Other	1001 (7.7)	320 (9.4)	266 (9.5)	181 (8.9)	
GINA Step, n (%)					
I	4875 (37.7)	1603 (46.9)	1277 (45.4)	752 (37.0)	<0.001
2	3035 (23.4)	764 (22.4)	578 (20.5)	463 (22.8)	
3	2700 (20.9)	564 (16.5)	513 (18.2)	425 (20.9)	
4	2330 (18.0)	488 (14.3)	448 (15.9)	391 (19.2)	
5	7 (0.1)	0 (0.0)	0 (0.0)	2 (0.1)	
BMI (kg/m <sup>2</sup> ), n (%)	n=10,245	n=2507	n=2303	n=1802	
Underweight (<18.5)	436 (4.3)	136 (5.4)	110 (4.8)	105 (5.8)	0.004
Normal (≥18.5–<22.9)	2361 (23.1)	554 (22.1)	586 (25.5)	412 (22.9)	
Overweight (≥23–<27.5)	3431 (33.5)	837 (33.4)	747 (32.4)	627 (34.8)	
Obese (≥27.5)	4017 (39.2)	980 (39.1)	860 (37.3)	658 (36.5)	
Smoking status, n (%)	n=10,388	n=2739	n=2523	n=1742	
Never smoked	9540 (91.8)	2390 (87.3)	2161 (85.7)	1330 (76.4)	<0.001
Ex-smoker	175 (1.7)	59 (2.2)	59 (2.4)	55 (3.2)	
Current smoker	673 (6.5)	290 (10.6)	303 (12.0)	357 (20.5)	

Table I Baseline Demographic and Clinical Characteristics

Notes: Patients were categorized into 4 cohorts based on their index year, from 2016 to 2019.

 $\label{eq:abbreviations: BMI, body mass index; GINA, Global Initiative for Asthma.$ 

by a gradual increase to 42.7% (n=868) in the 2019 cohort (p<0.001). Similarly, the proportion of patients at GINA Step 3 or above decreased from 38.9% (n=5037) in the 2016 cohort to 30.8% (n=1052) in the 2017 cohort, before increasing to 40.2% (n=818) in the 2019 cohort (p<0.001), whereas the proportion of current smokers increased from 2016 (6.5% [n=673]) to 2019 (20.5% [n=357]) (p<0.001) (Table 1).

There was a decreasing trend for comorbid pneumonia from 7.9% (n=1023) in 2016 to 4.8% (n=98) in 2019 (p<0.001) during the baseline period, while comorbid hypertension and heart failure decreased from 52.3% (n=6772) and 3.0% (n=390) in 2016 to 29.0% (n=991) and 1.8% (n=62) in 2017, respectively, before remaining relatively stable from 2017 to 2019 (p<0.001 for all) (Supplementary Table 4). Mean (SD) blood EOS counts were similar across index years during the baseline period (0.30–0.31 [0.29–0.47] × 10<sup>9</sup> cells/L; p<0.999) (Supplementary Table 5).

### Asthma Medication Use

The most common asthma medication during the baseline period changed with increasing index year; inhaled corticosteroid (ICS) monotherapy was the most frequently prescribed medication in the 2016 cohort (32.8% [n=4252]) but was less frequently prescribed in the 2019 cohort (26.1% [n=531]), whereas the proportion of patients prescribed ICS/longacting  $\beta_2$ -agonist (LABA) therapy increased from 25.1% (n=3252) in the 2016 cohort to 33.3% (n=677) in 2019 (*p*<0.001) (Figure 1; Supplementary Table 6). Among patients receiving ICS monotherapy, the proportion receiving lowdose ICS increased between the 2016 and 2019 cohorts (71.6% vs 85.2%), while the proportion using medium-dose



■ ICS ■ ICS + LABA ■ ICS + LABA + montelukast ■ ICS + LABA + LAMA ■ ICS + montelukast

Figure I Inhaled medication use during the baseline period. Differences across index years per medication type were tested for significance using Chi-square tests: ICS + LABA, p<0.001; ICS + LABA + montelukast, p<0.001; ICS + LABA + LAMA, p<0.011; ICS + montelukast, p<0.236. Differences in ICS monotherapy use across index years were not assessed.

Abbreviations: ICS, inhaled corticosteroid; LABA, long-acting  $\beta_2$ -agonist; LAMA, long-acting muscarinic antagonist.

(24.2% vs 13.3%) or high-dose ICS decreased (3.3% vs 0.6%) (*p*<0.001 for all) (Supplementary Table 6). Similar trends for high- and low-dose ICS were also reported for ICS/LABA (Supplementary Table 6).

The proportion of patients prescribed OCS (maintenance or burst) showed a significant decreasing trend from 27.3% (n=3533) in the 2016 cohort to 15.5% (n=316) in the 2019 cohort (p<0.001); similar results were seen for those receiving systemic corticosteroid (SCS) bursts defined as those comprising more than two 5- to 7-day SCS courses (2016: 1.3% [n=174]; 2019: 1.0% [n=21]; p<0.001) (Table 2). The mean SCS dose and proportion of patients prescribed rescue therapy remained broadly similar across cohorts.

	2016 (N=12,947)	2017 (N=3419)	2018 (N=2816)	2019 (N=2033)	p value
OCS prescription (prednisolone), n (%)	3533 (27.3)	747 (21.9)	626 (22.3)	316 (15.5)	<0.001
SCS bursts*, n (%)	174 (1.3)	43 (1.3)	59 (2.1)	21 (1.0)	<0.001
SCS dose (mg, prednisolone), mean (SD)	32.17 (9.15)	32.39 (9.74)	32.67 (10.24)	32.85 (10.17)	0.432
Rescue therapy prescription <sup>†</sup> , n (%)	2156 (16.7)	575 (16.8)	465 (16.5)	333 (16.4)	0.976

Notes: \*SCS burst prescriptions were defined as those comprising  $\geq$ 2 5- to 7-day SCS courses; <sup>†</sup>salbutamol (SABA, 10 puffs) or ipratropium bromide (LAMA, 4 puffs) recorded in EMRs from SHPs.

**Abbreviations:** EMR, electronic medical record; LAMA, long-acting muscarinic antagonist; OCS, oral corticosteroid; SABA, short-acting  $\beta_2$ -agonist; SCS, systemic corticosteroid; SD, standard deviation; SHP, SingHealth Polyclinic.

# Exacerbations

The mean (SD) number of exacerbations per patient during follow-up showed a significantly decreasing trend from 2016 to 2019, with 1.91 (3.11) exacerbations in the 2016 cohort compared with 0.89 (2.07) in the 2019 cohort (p<0.001) (Figure 2). The proportion of patients with  $\geq$ 2 exacerbations during follow-up also decreased between 2016 and 2019 (48.6% [n=6295] vs 19.5% [n=396]), while the proportion of patients with no exacerbations increased between these cohorts (2016: 46.3% [n=5994]; 2019: 73.3% [n=1490]) (Figure 2).

# HCRU

HCRU during the follow-up period from 2016 to 2019 is presented in Figure 3. The mean (SD) number of primary care (SHP) outpatient visits per patient decreased from 1.89 (1.66) in 2016 to 1.19 (1.37) in the 2017 cohort, before increasing to 1.63 (1.87) in the 2019 cohort (p<0.001). The number of speciality care (SOC) outpatient visits showed an increasing trend from 2016 (0.28 [0.85]) to 2019 (0.55 [1.2]; p=0.0015). The mean (SD) number of ED visits was highest in the 2016 cohort (0.21 [0.75]), before decreasing and remaining relatively stable from 2017 to 2019 (0.17–0.18 [0.60–0.65]; p<0.001). The proportion of patients with no ED visits increased between the 2016 and 2019 cohorts (70.5% [n=9124] vs 86.5% [n=1758]), whereas the proportion with  $\geq$ 1 visit decreased from 2016 to 2019 (29.5% [n=3823] vs 13.5% [n=275]) (Figure 3).

# Asthma Symptom Control

Asthma symptom control generally improved with increasing index year (Figure 4). During the baseline period, the mean (SD) ACT score increased from 18.38 (4.93) in the 2016 cohort to 19.87 (4.56) in the 2019 cohort (p<0.001); mean (SD)



#### Number of exacerbations during the follow-up period

Exacerbations per year, n (%)	2016	2017	2018	2019
	(N=12,947)	(N=3419)	(N=2816)	(N=2033)
0	5994 (46.3)	1883 (55.1)	1724 (61.2)	1490 (73.3)
1	403 (3.1)	122 (3.6)	100 (3.6)	86 (4.3)
2	255 (2.0)	99 (2.9)	93 (3.3)	61 (3.0)
>2	6295 (48.6)	1315 (38.5)	899 (31.9)	396 (19.5)

Figure 2 Exacerbations during the follow-up period. Abbreviation: SD, standard deviation.



Figure 3 HCRU during the follow-up period.

Abbreviations: ED, emergency department; HCRU, healthcare resource utilization; SD, standard deviation; SHP, SingHealth Polyclinic; SOC, specialist outpatient clinic.

ACT scores also increased between the 2016 and 2019 cohorts during follow-up (18.34 [4.23] vs 21.07 [3.51]) (p<0.001) (Figure 4). Asthma symptom control levels also generally improved after index, with mean (SD) ACT scores increasing from baseline to the follow-up period in all cohorts except for 2016, which remained stable (Figure 4). In addition, the proportion of patients with not well-controlled asthma (ACT score <20) decreased between baseline and follow-up in all cohorts except for the 2016 cohort, which increased from 51.6% (n=112) to 53.3% (n=171) (Figure 4).

### Vaccinations and Asthma Counselling

Vaccination rates during the follow-up period decreased between the 2016 and 2019 cohorts, with a greater proportion of patients in 2016 receiving an influenza (43.2% [n=5598] vs 33.4% [n=679]), pneumococcal PCV13 (16.5% [n=2139] vs 9.8% [n=199]; p<0.001), or pneumococcal PPSV23 vaccination (13.7% [n=1775] vs 7.6% [n=154]; p<0.001) versus in 2019 (Supplementary Table 7).

Documentation of asthma counselling at least once in EMRs increased significantly from 16.5% (n=2131) in 2016 to 33.6% (n=682) in 2019 (p<0.001) (Table 3). Similarly, there was a significant increase in the proportion of patients with  $\geq$ 1 documentation of a WAAP from 66.5% (n=8612) in 2016 to 90.8% (n=1846) in 2019 (p<0.001) (Table 3).

# Discussion

This retrospective study is the first in Singapore to describe real-world characteristics of patients with asthma across a 4-year period (2016–2019), using both primary and speciality care data collected from EMRs. The proportion of patients aged  $\geq$ 54 years during the baseline period decreased from 2016 to 2017, before gradually increasing in the 2018



Figure 4 Asthma symptom control during the baseline and follow-up periods. Abbreviations: ACT, Asthma Control Test; SD, standard deviation.

and 2019 cohorts. Health outcomes generally improved from 2016 to 2019, with reductions in exacerbations and ED visits, and improvements in asthma symptom control in the 2019 cohort compared with the 2016 cohort.

Previous data from patients with severe, life-threatening asthma presenting at four public hospitals in Singapore from 2011 to 2015 show that 17% of patients had no regular asthma follow-up, and almost a third had no prior ICS controller medication,<sup>14,15</sup> suggesting suboptimal asthma care before 2016. In this study, we observed a greater proportion of patients at GINA Steps 3, 4, and 5 in the 2019 cohort versus the 2016 cohort. Moreover, between 2016 and 2019, the proportion of patients receiving asthma counselling more than doubled and the proportion receiving a WAAP increased from 67% to more than 90%. In addition, the most frequently prescribed asthma medication changed from ICS monotherapy in 2016 to ICS/LABA dual therapy in 2019. Collectively, these findings likely reflect improved patient management in recent years, where therapy is stepped-up appropriately for patients with uncontrolled asthma, potentially leading to the reduction in exacerbations and ED visits, and improvements in asthma symptom control reported between 2016 and 2019. The apparent improved care in recent cohorts may be due to wider adoption of asthma management

Table 3 Asthma	Counselling and	WAAP During	the Follow-Up Period
----------------	-----------------	-------------	----------------------

	2016 (N=12,947)	2017 (N=3419)	2018 (N=2816)	2019 (N=2033)	þ value
$\geq$ I documentation of asthma counselling in EMRs, n (%)	2131 (16.5)	567 (16.6)	733 (26.0)	682 (33.6)	<0.001
$\geq$ I documentation of a WAAP, n (%)	8612 (66.5)	2007 (58.7)	2118 (75.2)	1846 (90.8)	<0.001

Abbreviations: EMR, electronic medical record; WAAP, written asthma action plan.

guidelines and greater awareness among clinicians of the importance of achieving asthma control earlier in the treatment course.

Between the 2016 and 2019 index dates, an increasing trend in the frequency of SOC visits was observed. In contrast, the average number of primary care SHP visits was highest in the 2016 cohort and dropped considerably in the 2017 cohort. This reduction may have been influenced by the considerably larger patient population in the 2016 cohort versus 2017, 2018, or 2019 cohorts. Similar to the trend observed with SOC visits, from 2017 to 2019 the frequency of SHP visits also gradually increased, suggesting in later index years, patients were more often visiting their physicians in any care setting. As a result, patients likely received closer monitoring and more appropriate treatments, potentially leading to the improved outcomes observed at later index years in this study.

These improved outcomes possibly reflect the impact of local initiatives in Singapore, such as the "Towards Zero Asthma Deaths" campaign, which aimed to improve awareness of the importance of adhering to asthma management guidelines.<sup>16</sup> In addition, several ICS/LABA combinations were included on the Singapore Ministry of Health's (MOH) list of subsidized drugs during this period.<sup>17</sup> Improved access and affordability of ICS/LABA may account for the shift in prescription patterns from 2016 to 2019, affording patients the opportunity to receive more appropriate treatments. The findings from this study may be useful in providing further evidence to physicians in Singapore, particularly those in primary care, that adhering to asthma management guidelines can lead to improved patient outcomes.

The data reported here indicate there are still aspects of clinical practice and asthma management in Singapore that warrant improvement. Although the proportion of patients prescribed OCS or receiving SCS bursts decreased from the 2016 cohort to the 2019 cohort, the mean dose of SCS and proportion of patients prescribed rescue therapy remained relatively stable between cohorts. This suggests that despite greater focus on GINA-recommended treatment step-up with increasing year, there is an issue of high OCS/SCS and rescue therapy prescribing. These findings are notable given that short-acting  $\beta_2$ -agonist reliever use has been associated with reduced asthma control and increased risk of exacerbations,<sup>18–20</sup> while OCS/SCS use has been shown to result in greater asthma burden and increased risk of adverse effects.<sup>21–25</sup> These observations may also be a consequence of limited access and affordability of asthma biologics (GINA Step 5) in Singapore due to lack of government reimbursement.<sup>6,26</sup> Improving affordability and access to biologics in patients with severe asthma will likely reduce OCS burden and further reduce HCRU as shown in clinical trials<sup>27,28</sup> and real-world large registry data.<sup>6,26</sup>

In the present study, ACT scores showed an increasing trend across index years, suggestive of improved disease management over time. However, approximately 1 in 4 patients in the 2019 cohort continued to have asthma that was not well controlled during the follow-up period. Notably, although ICS/LABA was the most prescribed asthma medication in the 2019 cohort, more than a quarter of patients still received ICS monotherapy. This suggests that an unmet need remains for optimized treatment for patients in this study population. In line with 2023 GINA recommendations, the Singapore MOH Agency for Care Effectiveness guidelines for asthma recommend a step-wise treatment approach, whereby ICS is the mainstay of asthma management, and treatment decisions are based on long-term assessment of patients' symptoms, risk of exacerbations and hospital admissions.<sup>29</sup> For patients with uncontrolled asthma despite low-dose ICS/LABA, addition of a LABA at Step 3 is recommended at Step 4.<sup>29</sup> Raising awareness among clinicians about treatment optimization may help to promote best practice and improve patient outcomes, and further research could be done to assess the impact that these more recent guidelines are having on asthma management and patient outcomes, and to understand the barriers to patients and clinicians carrying out the guideline recommendations.

The inclusion of patients with ACO in this study was intended to provide a comprehensive view of real-world asthma management in Singapore, recognizing the frequent coexistence of asthma and COPD in clinical practice. While the classification of ACO is not universally standardized and may introduce some variability into the study's findings, including these patients was considered crucial to accurately capture the complexities of managing asthma in patients with overlapping characteristics of COPD. This approach enhances the study's applicability to real-world settings, where such comorbid conditions are often encountered.

This real-world, longitudinal study addresses knowledge gaps regarding characteristics, change in care patterns, and impact on health outcomes among patients with asthma in Singapore over a 4-year period, and provides valuable insight

from a population health perspective. The study draws from a comprehensive dataset including primary and speciality care data, which have previously been scarcely reported in the literature, as well as patient-level EMR data, which have not previously been reported for South-East Asia. However, the study has several limitations that should be noted. There were considerably more patients in the 2016 cohort versus the other cohorts due to the inclusion of pre-existing and newly diagnosed asthma cases in 2016, which may potentially have impacted the trends observed across cohorts. The intent of the study was to assess broader trends in asthma management and outcomes at the population level, reflecting changes in clinical practice and policy implementations over the years, rather than to compare identical cohorts across different years. As a result, the comparison of different cohorts by year may involve patient populations with inherent differences due to variations in demographic factors, disease severity, or treatment approaches over time, which introduces a limitation in interpreting the observed trends. The lag time or incomplete data capture of asthma symptom control may be due to a lack of integration of ACT results into the main SCDM platform following practice changes during the observation period. Compounded by a multiplicity of possible ACT reporting systems, the asthma symptom control trends observed in this study may not be representative of the entire cohort. Moreover, medication data are based on prescriptions rather than consumption and may instead have been prescribed as standby medication, rather than for immediate use. A further limitation is that the dataset only included patients treated in SHP or SGH outpatient clinics. Therefore, data for patients who visited other public or private clinics<sup>30</sup> were not captured. Additionally, patients in primary care may experience severe exacerbations or be referred to clinics not included in the SCDM: these events were not captured, therefore, the asthma severity of these patients may be underestimated. Furthermore, the study's design was not intended to track individual patients over extended periods due to the database structure, which limited follow-up to 1 year post index diagnosis. While this reflects the pragmatic nature of the analysis, it restricts the ability to capture longterm changes in asthma management and outcomes. Finally, the SCDM database primarily focuses on clinical data related to asthma management, without detailed socioeconomic information. Future studies will need to incorporate socioeconomic data to provide a more comprehensive analysis of asthma management and outcomes.

### Conclusions

This is the first study in Singapore to describe the real-world characteristics and outcomes of an incident asthma population over a 4-year period in patients managed in both primary and tertiary care settings. These results show overall improvements in outcomes for patients with asthma with increasing index year from 2016 to 2019, including reduced exacerbations and ED visits, improved asthma symptom control, and more asthma counselling. This provides evidence that policy changes and implementation such as improving disease awareness and adoption of guideline recommendations is successful in improving quality of care and outcomes at a population level for people with asthma in Singapore.

### **Abbreviations**

ACO, asthma-COPD overlap; ACT, Asthma Control Test; COPD, chronic obstructive pulmonary disease; ED, emergency department; EMR, electronic medical record; EOS, eosinophil; GINA, Global Initiative for Asthma; HCRU, healthcare resource utilization; ICS, inhaled corticosteroid; LABA, long-acting  $\beta_2$ -agonist; MOH, Ministry of Health; OCS, oral corticosteroid; SCDM, SingHealth COPD and Asthma Data Mart; SCS, systemic corticosteroid; SD, standard deviation; SDG-CARE, SingHealth-Duke-GSK COPD and Asthma Real-World Evidence; SGH-SOC, Singapore General Hospital respiratory medicine specialist outpatient clinic; SHP, SingHealth Polyclinic; US\$, United States dollar; WAAP, written asthma action plan.

### **Data Sharing Statement**

Data from the SingHealth COPD and Asthma Data Mart (SCDM) may be available on reasonable request with deidentified patient data, subject to data owners, PIs, HODs and other stakeholders' approval, from the corresponding author.

# **Ethics Approval and Informed Consent**

This study complied with all applicable laws regarding subject privacy. No direct patient contact or primary collection of individual human participant data occurred. Study results are in tabular form and presented as aggregate analyses that omit patient identification, therefore, informed consent, ethics committee or International Review Board approval were not required; waiver of appropriate consent can be found under the Human Biomedical Research Act (HBRA) Fifth Schedule, Part 2, Section 3 (individually identifiable health information or human biological material obtained or compiled before, on and/or after 1 Nov 2017).

# Acknowledgments

Editorial support, in the form of preparation of the first draft based on input from all authors, and collation and incorporation of author feedback to develop subsequent drafts, was provided by Ben Usher, PhD, of Fishawack Indicia Ltd, UK, part of Avalere Health, and was funded by GSK.

The authors would like to thank Mr See Wei Qiang (HSRC) for his help with project management and the Singapore General Hospital Airway coordinators (Ms Karen Tan, Ms Yvonne Tiang, Ms Noor Syifa Binte Shamsuddin, Ms Sherine Lim Shu Gim and Ms Cheong Ai Wei) for their help with clinical service and data entry.

# **Author Contributions**

All authors made a significant contribution to the work reported; MSK, SSWL, NCT and CML contributed to the conception and design of the study and data interpretation. MEHO contributed to the data analysis and interpretation. XX and DBM contributed to the conception and design of the study and data interpretation. JTW contributed to the acquisition of data and data analysis. PR and RM contributed to data interpretation. All authors took part in drafting, revising or critically reviewing the article; gave final approval of the version to be published; have agreed on the journal to which the article has been submitted; and agree to be accountable for all aspects of the work.

# Funding

This study and the SGD-CARE collaboration were funded by GSK (GSK study 218080).

# Disclosure

MSK reports grant support from AstraZeneca and honoraria for lectures and advisory board meetings paid to Singapore General Hospital from GSK, AstraZeneca, Novartis, Sanofi and Boehringer Ingelheim, and Roche outside of the submitted work. XX is an employee of GSK and has financial equities in GSK. PR was an employee of GSK at the time of the study. RM is an employee of GSK. SSWL, JTW, MEHO, DBM, NCT and CML declare no competing interests for this work.

# References

- 1. Wang Z, Li Y, Gao Y, et al. Global, regional, and national burden of asthma and its attributable risk factors from 1990 to 2019: a systematic analysis for the global burden of disease study 2019. *Respir Res.* 2023;24(1):169. doi:10.1186/s12931-023-02475-6
- 2. Jeyagurunathan A, Abdin E, Shafie S, et al. Asthma prevalence and its risk factors among a multi-ethnic adult population. *Yale J Biol Med.* 2021;94 (3):417–427. doi:10.1164/rccm.200907-1035OC
- 3. Singapore Ministry of Health. National health survey 2010. Updated November 2011. Available from: https://www.moh.gov.sg/resources-statistics /reports/national-health-survey-2010. Accessed October 24, 2023.
- 4. Wong QYA, Lim JJ, Ng JY, et al. An updated prevalence of asthma, its phenotypes, and the identification of the potential asthma risk factors among young Chinese adults recruited in Singapore. *World Allergy Organ J.* 2023;16(3):100757. doi:10.1016/j.waojou.2023.100757
- 5. Thompson PJ, Salvi S, Lin J, et al. Insights, attitudes and perceptions about asthma and its treatment: findings from a multinational survey of patients from 8 Asia-Pacific countries and Hong Kong. *Respirology*. 2013;18(6):957–967. doi:10.1111/resp.12137
- 6. Lim GN, Allen JC, Tiew PY, Chen W, Koh MS. Healthcare utilization and health-related quality of life of severe asthma patients in Singapore. J Asthma. 2023;60(5):969–980. doi:10.1080/02770903.2022.2114086
- 7. Finkelstein EA, Lau E, Doble B, Ong B, Koh MS. Economic burden of asthma in Singapore. *BMJ Open Respir Res.* 2021;8(1). doi:10.1136/bmjresp-2020-000654
- 8. Quek JS, Tang WE, Chen E, Smith HE. Understanding the journeys of patients with an asthma exacerbation requiring urgent therapy at a primary care clinic. *BMC Pulm Med.* 2022;22(1):231. doi:10.1186/s12890-022-02024-9

- 9. Lam SSW, Fang AHS, Koh MS, et al. Development of a real-world database for asthma and COPD: the SingHealth-Duke-NUS-GSK COPD and Asthma real-world evidence (SDG-CARE) collaboration. *BMC Med Inform Decis Mak.* 2023;23(1):4. doi:10.1186/s12911-022-02071-6
- 10. Singapore Department of Statistics (DOS). Population trends 2023. 2023. Available from: www.singstat.gov.sg. Accessed November 20, 2024.
- 11. SNOMED International. Available from: https://www.snomed.org/. Accessed March 26, 2024.
- 12. Global Initiative for Asthma. Global strategy for asthma management and prevention. Available from: https://ginasthma.org/wp-content/uploads/ 2023/04/GINA-2015-main-report-final-WMSA.pdf. Accessed December 7, 2023.
- 13. Centers for Disease Control and Prevention. International classification of diseases, tenth revision, clinical modification (ICD-10-CM). [updated June 29, 2023]. Available from: https://www.cdc.gov/nchs/icd/icd-10-cm.htm. Accessed March 26, 2024.
- 14. Yii A, Puah SH, Lim HF, et al. A national audit of severe life-threatening asthma in Singapore. Eur Respir J. 2018;52(Suppl. 62):PA3957.
- 15. Yii ACA, Tay TR, Puah SH, et al. Blood eosinophil count correlates with severity of respiratory failure in life-threatening asthma and predicts risk of subsequent exacerbations. *Clin Exp Allergy*. 2019;49(12):1578–1586. doi:10.1111/cea.13465
- 16. Chai HZ, Koh SM. Zero tolerance towards asthma deaths in Singapore: role of the family doctor. Singapore Family Physician. 2018;4(4):10.
- 17. Singapore Ministry of Health. List of subsidised drugs. [updated August 1, 2023]. Available from: https://www.moh.gov.sg/healthcare-schemes-subsidised-drug-list. Accessed October 24, 2023.
- 18. Bateman ED, Price DB, Wang HC, et al. Short-acting β 2 -agonist prescriptions are associated with poor clinical outcomes of asthma: the multicountry, cross-sectional SABINA III study. Eur Respir J. 2022;59(5):2101402. doi:10.1183/13993003.01402-2021
- Global Initiative for Asthma. Global strategy for asthma management and prevention. Available from: https://ginasthma.org/2023-gina-main-report/. Accessed October 24, 2023.
- 20. Nwaru BI, Ekstrom M, Hasvold P, Wiklund F, Telg G, Janson C. Overuse of short-acting β 2 -agonists in asthma is associated with increased risk of exacerbation and mortality: a nationwide cohort study of the global SABINA programme. *Eur Respir J.* 2020;55(4):1901872. doi:10.1183/13993003.01872-2019
- Dhar R, Rhee CK, Perng DW, et al. The burden of systemic corticosteroid use in asthma management in Asia. *Respirology*. 2023;28(8):744–757. doi:10.1111/resp.14533
- Al Efraij K, Johnson KM, Wiebe D, Sadatsafavi M, FitzGerald JM. A systematic review of the adverse events and economic impact associated with oral corticosteroids in asthma. J Asthma. 2019;56(12):1334–1346. doi:10.1080/02770903.2018.1539100
- Zeiger RS, Schatz M, Li Q, Chen W, Khatry DB, Tran TN. Burden of chronic oral corticosteroid use by adults with persistent asthma. J Allergy Clin Immunol Pract. 2017;5(4):1050–1060e9. doi:10.1016/j.jaip.2016.12.023
- 24. Lee JH, Kim HJ, Park CS, et al. Clinical characteristics and disease burden of severe asthma according to oral corticosteroid dependence: real-world assessment from the Korean Severe Asthma Registry (KoSAR). *Allergy Asthma Immunol Res.* 2022;14(4):412–423. doi:10.4168/ aair.2022.14.4.412
- 25. Sullivan PW, Ghushchyan VH, Globe G, Schatz M. Oral corticosteroid exposure and adverse effects in asthmatic patients. *J Allergy Clin Immunol*. 2018;141(1):110–116e7. doi:10.1016/j.jaci.2017.04.009
- 26. Tan LE, Tan WHG, Aziz MIA, et al. Assessing the cost-effectiveness of mepolizumab as add-on therapy to standard of care for severe eosinophilic asthma in Singapore. J Asthma. 2022;59(1):189–199. doi:10.1080/02770903.2020.1837158
- 27. Ortega HG, Liu MC, Pavord ID, et al. Mepolizumab treatment in patients with severe eosinophilic asthma. N Engl J Med. 2014;371 (13):1198–1207. doi:10.1056/NEJMoa1403290
- 28. Castro M, Corren J, Pavord ID, et al. Dupilumab efficacy and safety in moderate-to-severe uncontrolled asthma. N Engl J Med. 2018;378 (26):2486–2496. doi:10.1056/NEJMoa1804092
- 29. Agency for Care Effectiveness. Asthma optimising long-term management with inhaled corticosteroid. Updated October 15, 2020. Available from: https://www.ace-hta.gov.sg/healthcare-professionals/ace-clinical-guidances-(acgs)/details/asthma-optimising-long-term-management-with-inhaled-corticosteroid. Accessed October 24, 2023.
- 30. Fang Y, Soljak M, Tan SLL, Peckham S, Tan TL, Smith HE. General practitioners' views on retaining Singapore's primary care doctors: a crosssectional survey and qualitative analysis. *BMC Prim Care*. 2022;23(1):168. doi:10.1186/s12875-022-01774-z

Pragmatic and Observational Research



Publish your work in this journal

Pragmatic and Observational Research is an international, peer-reviewed, open access journal that publishes data from studies designed to reflect more closely medical interventions in real-world clinical practice compared with classical randomized controlled trials (RCTs). The manuscript management system is completely online and includes a very quick and fair peer-review system. Visit http://www.dovepress.com/ testimonials.php to read real quotes from published authors.

Submit your manuscript here: http://www.dovepress.com/pragmatic-and-observational-research-journal